Author’s response to reviews

Title: Do parents’ support behaviours predict whether or not their children get sufficient sleep?: A cross-sectional study

Authors:

Evelyn Pyper (pyper.evelyn@gmail.com)
Daniel Harrington (Daniel.Harrington@oahpp.ca)
Heather Manson (Heather.Manson@oahpp.ca)

Version: 1 Date: 17 Oct 2016

Author’s response to reviews:

Dear Gerard Clunn,

Thank you for your feedback regarding the statement of consent. Based on your comments, we have made this section more specific. In the paper, it now states, "Informed consent to participate in the study was obtained from all participants before proceeding with the telephone survey. Given that the method of data collection was a computer assisted telephone interview, written consent was not feasible, and thus, verbal consent was obtained and documented by the interviewer."

We agree that the above explanation is much more clear and informative than what was previously written.

Kind regards,

Evelyn Pyper